Author’s response to reviews

Title: Flare or foe? - Mycobacterium marinum infection mimicking rheumatoid arthritis tenosynovitis: case report and literature review

Authors:

Nils Schubert (nils.schubert1@med.uni-goettingen.de)

Tillmann Schill (tillmann.schill@med.uni-goettingen.de)

Marlene Plüß (marlene.pluess@med.uni-goettingen.de)

PETER KORSTEN (peter.korsten@med.uni-goettingen.de)

Version: 4 Date: 08 Jan 2020

Author’s response to reviews:

Dear Editor,

- Figure 4 is now referenced in the text (page 7: "Figure 4 summarizes the published cases.")
- CARE checklist has been removed
- clean version is being submitted

Kind regards,

Peter Korsten